

HHS Public Access

Author manuscript

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

Published in final edited form as:

Am Econ Rev. 2020 December; 110(12): 3836–3870. doi:10.1257/aer.20191191.

Screening and Selection: The Case of Mammograms

Liran Einav,

Department of Economics, Stanford University, and the National Bureau of Economic Research

Amy Finkelstein,

Department of Economics, Massachusetts Institute of Technology, and the National Bureau of Economic Research

Tamar Oostrom,

Department of Economics, Ohio State University

Abigail Ostriker,

Department of Economics, Massachusetts Institute of Technology

Heidi Williams^{*}

Department of Economics, Stanford University, and the National Bureau of Economic Research

Abstract

We analyze selection into screening in the context of recommendations that breast cancer screening start at age 40. Combining medical claims with a clinical oncology model, we document that compliers with the recommendation are less likely to have cancer than younger women who select into screening or women who never screen. We show this selection is quantitatively important: shifting the recommendation from age 40 to 45 results in three times as many deaths if compliers were randomly selected than under the estimated patterns of selection. The results highlight the importance of considering characteristics of compliers when making and designing recommendations.

Whether and when to recommend screening for potential diseases is a highly controversial and evolving policy area, with active academic research.¹ Much of the debate—both in public policy and in academia—centers on the causal impact of screening for a typical individual covered by the recommendation. Estimating this causal impact is challenging for several well-known reasons. First, there are the usual challenges to causal inference. Second, many of the potential costs and benefits of screening are difficult to measure and to monetize.² In this paper, we highlight another important—and, we believe, overlooked—challenge in analyzing and designing screening recommendations: the typical individual

leinav@stanford.edu.

¹For example, Welch, Schwartz, and Woloshin (2011) argue that although many medical conditions—such as high blood pressure, elevated blood glucose levels, low bone density, and high cholesterol—benefit from treatment, there has been a trend over time towards widespread use of medical screening tests and increasingly low diagnostic thresholds that recommend treating patients for whom the benefits from treatments are quite small. By contrast, Maciosek et al. (2010) review these same screening efforts and conclude that they save a large number of lives at relatively low cost. ²The costs and benefits of screening include monetary costs, clinical outcomes, discomfort from unnecessary procedures, and

²The costs and benefits of screening include monetary costs, clinical outcomes, discomfort from unnecessary procedures, and psychological effects induced by the screening process, including pre-screening apprehension and anxiety due to false positives (e.g., Brett et al. 2005; Nelson et al. 2009; Welch and Passow 2014; Ong and Mandl 2015; Welch 2015).

covered by a recommendation may be very different from the typical individual who responds to the recommendation. As a result, the estimated impact of screening for a randomly-selected individual may be quite different from the impact for an affected individual.

We explore this distinction in the context of the current controversy over whether to recommend annual mammograms for women starting at age 40. Results from randomized trials have consistently failed to show statistically significant mortality benefits of mammograms for women in their 40s. In 2009, these results prompted the US Preventive Services Task Force (USPTF) to change its recommendation for routine mammograms to begin at age 50 rather than at age 40. This change generated substantial public controversy (Kolata 2009; Saad 2009; Berry 2013).

This debate has focused on the costs and benefits of mammograms for typical ("averagerisk") 40-year-old women, with little attention paid to what types of women respond to a screening recommendation and whether the costs and benefits for them may differ from the average woman. To investigate the type of women who respond, we draw on two primary data sources. The first is insurance claims data on mammogram choices and their results (negative, false positive, or true positive) for privately-insured women aged 35–50 from the Health Care Cost Institute (HCCI). The second is cancer registry data, from the National Cancer Institute's Surveillance, Epidemiology, and End Results (SEER) database, on the size and stage of detected tumors for women aged 35–50 who were diagnosed with breast cancer. We supplement some of the descriptive analyses with additional information from the Behavioral Risk Factor Surveillance System Survey (BRFSS), which allows us to observe additional health behaviors and demographics of women who do and do not receive mammograms at various ages.

The visual evidence shows sharp and pronounced changes in behavior and outcomes at age 40. There is a 25-percentage-point jump in the annual mammogram rate at age 40, from 10 percent to 35 percent of women. We then compare characteristics of the women who respond to the recommendation for a mammogram (i.e. "compliers" in the terminology of Angrist, Imbens, and Rubin 1996) to characteristics of always-takers (i.e. women who choose mammograms even in the absence of the recommendation, which is before age 40). We find that compliers have a lower incidence of cancer than always-takers: there is a roughly 30 percent decline (from 0.84% to 0.56%) in the share of screened women diagnosed with cancer (i.e. true positives) at age 40. Given the high rate of false positivesabout 90 percent of initial positive mammograms turn out to be false positives—the sharp increase in the mammogram rate at age 40 translates into a substantial increase in the number of women experiencing false positives, from about 10 per thousand women to about 40 per thousand women. This is consistent with false positives being a key concern that motivated moving the recommended age of beginning mammography from 40 to 50 (Nelson et al. 2009). Moreover, among those diagnosed with cancer, the registry data show a sharp decline in the average tumor's stage and size starting at age 40, compared to earlier ages. For example, the share of detected tumors that are in a late stage (invasive tumors) as opposed to early stage (in-situ tumors) falls by about 6 percentage points (or 7 percent) at age 40.

These descriptive results indicate that women who respond to the recommendation for a mammogram have lower risk of cancer than those who seek mammograms in the absence of the recommendation. For non-cancer characteristics, we can also compare compliers to never-takers (women who do not get mammograms even once the recommendation is in effect). We find that, relative to never-takers, compliers are more likely to undertake other types of recommended preventive care, such as cervical cancer screening tests and flu shots. This pattern is consistent with findings that when a health behavior is recommended, those who comply with the recommendation tend to exhibit other positive health behaviors (Oster 2020). It also echoes the observation that women who comply with assignment to mammograms in an RCT setting are healthier than never-takers (Kowalski 2019).³

To assess the implications of these findings and to quantify costs and health outcomes under various counterfactual selection scenarios, we specify a model of mammogram demand that is a function of a woman's age, her (undiagnosed) cancer type (no cancer, in-situ, or invasive), and whether or not a mammogram is recommended at her age. We estimate the model by method of moments, using two key inputs. First, we leverage our data on the observed patterns of mammogram decisions and mammogram outcomes (specifically, cancer type) for women by age. Second, we bring in a clinical oncology model of the underlying rate of onset of breast cancer by age, as well as the cancer's clinical progression in the absence of detection and treatment.

The clinical model allows us to estimate the cancer characteristics of never-takers. In the absence of a clinical model, these cancer characteristics are inherently difficult (or impossible) to observe: cancer incidence is not observed in the non-screened population, and almost all detected cancer is treated immediately upon detection. The clinical model of breast cancer incidence and progression is drawn from a large-scale, coordinated project funded by the National Cancer Institute (NCI) involving seven different research groups (Clarke et al. 2006); since there is naturally some uncertainty about the underlying model, we confirm that our main findings are not sensitive to a range of alternative assumptions about the onset and distribution of cancer type by age.

The estimates from our model indicate that women who would select into mammograms in the absence of the recommendation ("always-takers") have much higher rates of both in-situ and invasive cancer than the general population. We refer to this as "positive selection" into mammograms (positive with respect to cancer incidence). However, our estimates indicate that the women who select into mammograms due to the recommendation ("compliers") are much less likely to have invasive cancer—and are no more likely to have in-situ cancer—than women who do not select into mammograms ("never-takers"). The relative degree of selection pre- and post- the age-40 recommendation is identified directly from our data; the clinical model of underlying cancer incidence is needed to assess whether the observed selection either pre- or post-age 40 is positive with respect to the underlying population, whose cancer incidence is not directly observed.

 $^{^{3}}$ Because the context is naturally quite different, one might expect selection into compliance with RCT assignment to be different than selection into compliance with a recommendation. Indeed, never-takers comprise only 5% of the population in the context of Kowalski (2019), but are approximately 60% in ours.

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

We apply our model and its estimates to illustrate how the nature of selection in response to the recommendation affects the impact of the recommendation. Specifically, we estimate that shifting the recommendation from age 40 to age 45 results in more than three times as many deaths-at similar cost savings-if we assume that compliers with the recommendation are randomly drawn from the population rather than drawn based on the estimated selection patterns. We view this as a particularly instructive counterfactual, since assuming that the women who respond are randomly drawn from the population is conceptually similar to using estimates of the impact of mammography from randomized experiments (with full compliance). Because in practice those who respond to the recommendation have a much lower rate of invasive cancer than the underlying population, the mortality cost of moving the recommendation to age 45 is lower than under random selection. Conversely, our model also illustrates that if it were feasible to target the recommendations to those with higher rates of cancer, the mortality cost of moving the recommendation from age 40 to 45 could be substantially larger than even the random selection assumption would imply. This is consistent with recent interest in reducing overdiagnosis by developing targeted, precision screening for women at higher risk (Elmore 2016; Esserman, Shieh, and Thompson 2009).

Our paper relates to several distinct literatures. Most narrowly, it speaks to the large body of work on mammograms, which we describe in the next section. But beyond the specific application of mammograms, it speaks to a broader health policy debate about whether and when to recommend medical screening tests (e.g., Welch, Schwartz, and Woloshin 2011). A central challenge that has limited empirical research on this topic is that—in the datasets typically available to researchers-the testing decision is observed but the outcome of the test is not. An attractive feature of our setting is that the outcome of the test (i.e. cancer incidence and type of cancer) is measurable both in claims data and in registry data. In this sense our analysis is similar in spirit to Abaluck et al. (2016), who are able to measure the outcome of imaging tests for pulmonary embolism in claims data, which they use to investigate whether and when that imaging test is being "overused". Both our paper and Abaluck et al. (2016) share a common feature with the racial profiling literature on stop and frisks (Anwar and Fang 2006; Persico 2009): the object of interest is only observed conditional on an action. This raises an empirical challenge for analyzing how the action (in our case, screening) relates to the underlying object of interest (in our case, the underlying incidence of cancer and cancer types). In our setting, we overcome this empirical challenge by combining two insights. First, the recommendation at age 40 serves as an exogenous source of variation in the screening rate, allowing us to estimate the cancer type of the marginal person affected by the recommendation. Second, the clinical oncology model of cancer incidence and growth allows us to use the observed moments (namely, outcomes conditional on screening under different regimes) to model outcomes under counterfactual regimes.

More broadly, our paper speaks to the value of complementing reduced-form estimates of causal effects with economic models of behavior, and particularly of selection. Reduced-form methods—both quasi-experimental and randomized experiments—aim to estimate causal effects by shutting down any endogenous choices. In practice, however, most policies involve an element of choice, so that the ultimate impact of the policy depends not only on

the distribution of causal treatment effects but also on which women select into treatment. In this sense, our paper relates broadly to the literature on Roy selection, or selection on gains. In the health care context specifically, Einav et al. (2013) emphasize that the impact on health care spending of offering a high-deductible health insurance plan may be very different than what would be estimated from random assignment of high-deductible plans across individuals, because the types of people who choose high-deductible plans can have very different health care utilization responses to cost sharing than a typical individual. Our analysis speaks to a similar issue, in the context of evaluating recommendations for disease screening.

The rest of the paper proceeds as follows. Section I summarizes the relevant institutional details of our empirical context (breast cancer and mammography), and describes the existing evidence regarding the effect of mammograms and of various policy interventions that are designed to increase mammography rates. Section II describes our data and presents descriptive results. Section III presents our model of mammogram choice and describes how we estimate it using the observed descriptive patterns together with a clinical oncology model. Section IV presents the model estimates and discusses their implications for the impact of changing the recommended age of beginning mammography under both observed and counterfactual selection patterns. The last section concludes by using our findings to speculate about possible policy implications more broadly.

I. Empirical context

A. Breast cancer

The earliest stages of breast cancer typically produce no symptoms and are not detectable in the absence of screening technologies.⁴ As breast cancer progresses, it can spread within the breast, to adjacent tissues, to adjacent lymph nodes, and to distant organs (known as metastases). In clinical settings, tumors are classified according to the size of the tumor, the extent to which it has spread to lymph nodes, and whether it has metastasized. Public health research typically relies on a standardized classification—namely, the SEER classification system—which includes four stages: in-situ, local, regional, and distant; the last three stages are collectively referred to as "invasive" tumors.

Our analysis focuses on the distinction between in-situ and invasive tumors, a distinction that has been a key focus of the policy debate around mammography recommendations. Insitu refers to abnormal cells that have not invaded nearby tissues, instead remaining confined to the ducts or glands in which they originated. Some but not all in-situ tumors will become invasive. Expected survival time varies greatly by stage at diagnosis: women who are diagnosed with localized breast cancer are 99% as likely as cancer-free women to survive to 5 years after diagnosis, compared to 85% for regional breast cancer, and 27% for distant-stage breast cancer.⁵ Within a stage, survival also varies with tumor size. For example, among women with regional disease, 5-year survival (again, relative to comparable cancer-

 $^{^{4}}$ Unless otherwise noted, the discussion in this section draws from the American Cancer Society (2017a).

⁵These tabulations are drawn from US SEER cancer registry data from 2007–2013, as in American Cancer Society (2017a).

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

free women) is 95% for tumors smaller than 2 centimeters in diameter, 85% for tumors of 2-5 centimeters, and 72% for tumors greater than 5 centimeters.⁶

B. Mammography

Asymptomatic breast cancer can be detected by a mammogram, which is a low-dose X-ray procedure that allows visualization of the internal structure of the breast. If an abnormality is detected on a routine screening mammogram, the woman is typically called back in for a diagnostic mammogram and—if needed—a confirmatory biopsy (Cutler 2008; Hubbard et al. 2011). Once a diagnosis has been confirmed, the woman may undergo surgery to remove the tumor, in addition to other treatments which aim to reduce the risk of recurrence, such as radiation therapy, chemotherapy, hormone therapy, and/or targeted therapy.

Mammography is based on the theory of early detection of invasive cancer, rather than detection and removal of precancerous lesions (Humphrey et al. 2002). The efficacy of mammography is the subject of considerable debate. Mechanically, mammography is most beneficial if machines can detect tumors in their earliest stages, and if tumors (on average) rapidly become more difficult to treat the longer they go undetected. The benefits from mammography will be lower if a tumor is slow to advance from stage to stage, if mortality when treatment begins at a later stage is similar to when tumors are treated earlier, or if mammogram machines are unlikely to correctly identify tumors. In practice, because most women diagnosed with breast cancer are treated immediately upon detection, there is little information about the natural history of breast cancer tumors, making it difficult to know how an individual tumor would have progressed had it not been treated (Zahl, Maehlen, and Welch 2008). This complicates attempts to quantify the benefits of mammography.

In principle, the major potential health benefit of mammography is reduced mortality. However, in practice, randomized trials of the impact of mammograms on mortality have documented mixed results (Habbema et al. 1986; Alexander et al. 1999; Miller et al. 2000, 2002; Nyström et al. 2002; Bjurstam et al. 2003; Moss et al. 2006). There have been nine trials in total, with the first one dating back to the 1960s (Welch and Black 2010). Their estimates of relative risk reduction in breast-cancer mortality due to invitation to mammography range from 0% to 31% (Welch and Passow 2014), but many of these studies have lacked the statistical power to separately determine effects in different age groups (Humphrey et al. 2002). In particular, while most studies indicate that mammography reduces mortality among average-risk women over age 50, recent trials specifically designed to study mammography in younger women (aged 40–49) have estimated statistically insignificant reductions in breast-cancer mortality in this age group (Bjurstam et al. 2003; Moss et al. 2006).

The potential costs of mammography include financial, physical, and psychological costs. These costs arise from the initial screening, the frequent finding of false positives, and the treatment of cancers that would not have become clinically relevant in a woman's lifetime (often referred to as "over-diagnosis") (Jørgensen and Gøtzsche 2009). Some of these costs, such as the financial cost of a screening, are easy to quantify, while others are much more

⁶These tabulations are drawn from US SEER cancer registry data from 2000–2014, as in American Cancer Society (2017a).

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

difficult to estimate. Estimates of the rate of over-diagnosis of breast cancer (from both observational work and inferences from randomized control trials) range from less than 5% to more than 50% of diagnosed breast cancers (Zackrisson et al. 2006; Jørgensen and Gøtzsche 2009; Bleyer and Welch 2012; Oeffinger et al. 2015; Harding et al. 2015; Welch et al. 2016; Jørgensen et al. 2017).⁷

C. Age-specific mammogram recommendation and its impact

Several studies have combined the existing evidence to quantify the costs and benefits of mammograms (e.g., Welch and Passow 2014; Ong and Mandl 2015). For example, Welch and Passow (2014) estimate that for every 1,000 women aged 40–49 who undergo annual mammography for 10 years, 0.1–1.6 women will avoid dying from breast cancer, while 510–690 will have at least one false-positive result and up to 11 women will be over-diagnosed and (unnecessarily) treated. As the estimates of the costs and benefits of mammography have evolved, so have the recommendations by medical associations regarding which groups of women should receive mammograms, and how often.

In the 1980s, following the first randomized trials of routine mammography, the National Institutes of Health (NIH), the National Cancer Institute (NCI), and eleven other health care organizations issued recommendations for routine screenings of women over age 40 (Kolata 2009). These recommendations became the subject of controversy over time as more trials were published, and the US federal government subsequently reconsidered its position. In 1997, an NIH panel concluded that there was insufficient evidence to recommend routine screening for women in their 40s, a finding that was controversial (one radiologist described the finding as a "death sentence" for women (Taubes 1997)). After public pressure, the Senate encouraged an advisory board to reject that conclusion (Kolata 2009). In 2009, following the publication of experimental data that failed to show statistically significant mortality benefits of mammograms for women in their 40s, the US Preventive Services Task Force (USPSTF) recommended that women begin screening at age 50. Again, this conclusion generated backlash from patient advocacy groups like the American Cancer Society, which at the time recommended annual screening for women aged 40 and above (American Cancer Society 2018).⁸ This negative reaction was exacerbated by fears that the Affordable Care Act (ACA, then being drafted) would allow insurers to refuse to cover mammograms for younger women. The USPSTF stood by its recommendation, but a poll found that 84% of women aged 35-49 did not plan to follow the new recommendations, and the ACA was modified to mandate that insurers reimburse mammograms for women aged 40 and over (Saad 2009). Although in the last few years most patient advocacy organizations have begun to moderate their stances, the question of whether mammography should be recommended in the 40-49 age group remains controversial.

Importantly, both the academic literature and the policy debate over the costs and benefits of mammograms have primarily focused on the average impacts of mammograms at specific

 ⁷Selection into screening potentially (partially) explains the phenomenon of over-diagnosis, since it results in more diagnoses of low-risk tumors.
 ⁸The American Cancer Society currently recommends annual screening for women between ages 45–54 and screening every 2 years

^oThe American Cancer Society currently recommends annual screening for women between ages 45–54 and screening every 2 years for women 55 years and older (American Cancer Society 2018).

ages. For example, Welch and Passow (2014) extrapolate results from mammography RCTs to the entire population without considering selection effects. In contrast, our focus is on the characteristics of women whose decision to get a mammogram is influenced by the mammogram recommendation, and how their underlying cancer incidence and characteristics may differ from that of a randomly-selected woman in the population.

Several papers have examined the mammogram response to recommendations (Kadiyala and Strumpf 2011, 2016; Jacobson and Kadiyala 2017). Most closely related to our work on the selected response to mammogram recommendations is Kadiyala and Strumpf (2016), who document a sharp increase in self-reported mammograms at age 40 and estimate that most of the "newly detected" cancers are early-stage cancers. Also closely related is the work of Kim and Lee (2017) and Bitler and Carpenter (2016), who document that women who elect to receive mammograms in response to price reductions are in better health than those who get the mammogram even without the price reduction or those who don't get the mammogram even with the price reduction. Finally, Kowalski (2019) shows that the compliers in a Canadian mammography RCT are healthier on both cancer dimensions (i.e., rates of breast cancer) and non-cancer dimensions (e.g. body mass and smoking) than the never-takers.

II. Data and descriptive patterns

A. Data and variable construction

Our analysis of mammogram choices and outcomes focuses on women aged 35–50 and draws on two primary data sources. The first is claim-level data provided by the Health Care Cost Institute (HCCI), consisting of all claims paid by three large commercial insurers (Aetna, Humana, and UnitedHealthcare) from January 2008 through December 2012. Together, these three insurers represented about one-quarter of individuals under age 65 with commercial insurance (HCCI 2012). The data capture the billing-related information contained in the claims that these insurers pay out to medical providers; this includes the exact date and purpose of each claim, as well as the amount paid by the insurer and the amount owed out of pocket. The data also include a (masked) person identifier as well as the individual's birth year and gender.

The claim-level information in the HCCI data allow us to construct variables measuring whether an individual had a screening mammogram,⁹ whether the result was positive or negative, and whether a positive result was a true positive or false positive. Our coding of screening mammograms (hereafter "mammograms")—as well as their outcomes—broadly follows the approach of Segel, Balkrishnan, and Hirth (2017), which we cross-validated using Medicare claims data linked to cancer registry data (see Appendix A for more details).

The complete HCCI data contain about 28.7 million privately-insured women aged 25–64, and over 70 million woman-years. We limit the data to woman-years aged 35–50 who are covered continuously for at least three years between January 2008 and December 2012; we

 $^{^{9}}$ A "screening mammogram" is a routine test that is conceptually different –and coded differently in the data – from a "diagnostic mammogram," which would typically follow the emergence of a possible breast cancer symptom (such as a positive screening mammogram).

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

keep all the years of coverage except the first and last (since for every woman-year we need to observe the previous year to define screening mammograms and the subsequent year to measure outcomes). This results in about 7.4 million woman-years, and 3.7 million distinct women over the years from January 2009 to December of 2011.

The primary drawback of the HCCI data is that we are not able to observe information on a breast cancer diagnosis beyond its detection. To overcome this limitation of the HCCI data, we also analyze the National Cancer Institute's (NCI) Surveillance, Epidemiology, and End Results (SEER) database. This is an administrative, patient-level cancer registry of all cancer diagnoses in 13 US states, covering about one quarter of the US population (SEER 2019). We analyze all the breast cancer diagnoses in the data between 2000 and 2014 for women aged 35–50 at the time of diagnosis; this covers about 212,000 diagnoses. All cancer diagnoses are required to be reported, with data collected directly from the cancer patients' medical records at the time of diagnosis (rather than self reports).¹⁰ For each diagnosed cancer, the SEER data contain information about the size and stage of each tumor at diagnosis. They also contain basic demographics for the patient including age at time of diagnosis, race, and insurance coverage, as well as subsequent mortality information through December 2013.

In our HCCI sample, the average woman's age is 43 and 27% of woman-years are under 40. In the SEER data, because cancer risk increases with age, the average age at diagnosis is a bit higher (44.6) and only 13% of the SEER diagnoses occur in women under 40. In SEER, where we can observe race, slightly over three-quarters of the sample is white. And unlike the HCCI data where, by construction, everyone is privately insured, in the SEER data only 84% are privately insured, while 13% are on Medicaid.

Table 1 documents mammogram rates and test results in the HCCI data. About 30% of woman-years are associated with a mammogram. The vast majority (89.6%) of mammograms are negative, and another 9.7% are false positives. Only 0.7% are true positives. Among all woman-years with a mammogram, total (insurer plus out-of-pocket) health care spending in the 12 months starting from (and including) the mammogram averages \$5,000; while it is slightly higher (by about \$1,000) for those with a false positive, it is dramatically higher for those with true positives, averaging almost \$50,000. Out-of-pocket spending in the 12 months post-mammogram is about \$2,800 for women with a positive mammogram, compared to \$715 for women with a negative mammogram and \$950 for women with a false positive.

The SEER data provide more information on tumor stage and tumor size for the 212,000 true positives (i.e. diagnoses) we observe. Just over 15% are in-situ; the rest are invasive. Of the invasive, about 57% are localized, 38% are regional, and the remaining 5% are distant.

¹⁰See https://seer.cancer.gov/manuals/2018/SPCSM_2018_maindoc.pdf for more information. SEER registries are required to collect data on persons who are diagnosed with cancer and who, at the time of diagnosis, are residents of the geographic area covered by the SEER registry.

B. Mammograms and outcomes, by age

Figure 1 shows the age profile of annual mammogram rates in the HCCI data. Because we observe birth year, the mammogram rate at age, say, 40 is the share of women who got a mammogram in the year they turned 40. Between ages 39 and 41, the mammogram rate jumps by over 25 percentage points, from 8.9% to 35.2%. This pronounced jump in mammogram rates at age 40 has been previously documented in self-reported data (Kadiyala and Strumpf 2011, 2016).¹¹ One might be concerned that the existence of a recommendation for mammograms at age 40 could bias upward self-reports at that age. However, our analysis, which uses claims data, confirms a real change in mammogram rates that we estimate at age 40 in the HCCI data is very similar to what we estimate using self-reported data (from the Behavioral Risk Factor Surveillance System Survey, or BRFSS), although—consistent with prior work (Blustein 1995; Cronin et al. 2009)—we estimate lower mammogram rates at every age in claims data compared to self-reported data.

We examine the outcomes of these mammograms—negative, false positive, and true positive —by age in the HCCI data. As shown in Appendix Figure A.2, the vast majority (85 to 90 percent) of mammograms are negative, and almost all of the remainder are false positives; spending is much higher for true positives than false positives and negatives.

Figure 2a shows the share of mammograms that are true positive and false positive by age. Between ages 39 and 41, the share of true positives falls by one-third (from 0.84% to 0.56%). This indicates that the marginal women who choose to have a mammogram because of the screening recommendation at age 40 (i.e. "compliers") have lower underlying rates of cancer (i.e. true positive diagnoses) than those who choose to get screened at younger ages before the recommendation kicks in ("always-takers").

The share of mammograms that are false positive is generally declining smoothly in age because the probability of a false positive is higher for women with denser breast tissue, and density generally decreases with age (Susan G. Komen Foundation 2018). The exception is a small "spike" in false positives around age 40; this likely is attributable to the fact that the probability of a false positive mammogram is highest for a woman's first mammogram (American Cancer Society 2017b). Note, however, that while the share of mammograms that are false positive is trending fairly smoothly in age, the share of women experiencing a false positive rises considerably at age 40, since there is a 25-percentage-point increase in the share of women who have a mammogram. This is shown in Figure 2b: the share of women experiencing a false positive mammogram quadruples at age 40, from about 10 to 40 per thousand women.

Figure 3a documents the age profile of tumor type among all diagnoses in the SEER data. Between ages 39 and 41, the share of detected tumors that are in-situ (as opposed to invasive) rises by 6 percentage points, from 11.6 percent to 17.9 percent; this is consistent

¹¹Our data span the time period when the 2009 US Preventive Services Task Force changed its recommendation for routine mammograms to begin at age 50 rather than at age 40. Past analyses, such as Block et al. (2013), have documented that this appears to have had little affect on women's mammography behavior, which is not surprising given the substantial public controversy over this recommendation change.

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

with prior findings from Kadiyala and Strumpf (2016). The average size of a detected tumor falls by over 10 percent, from 27.3mm at age 39 to 24.4mm at age 41, although the pattern is less dramatic since detected tumor size is also falling (albeit less rapidly) at earlier ages.

Finally, Figure 3b documents 5-year mortality post-diagnosis in the SEER data by age of diagnosis, separately for tumors initially diagnosed as in-situ and invasive tumors. Mortality is almost three times higher for invasive tumors compared to in-situ tumors. For example, at age 40, the five-year mortality rate is 16.2% for invasive tumors compared to 4.5% for insitu tumors. However, the mortality rate is roughly flat by age within tumor type.

C. Who responds to the recommendation?

The preceding descriptive results from both the HCCI and SEER data suggest that the women brought into screening by the recommendation at age 40 have a lower cancer disease burden than those who sought screening prior to the age-40 recommendation. This manifests in lower rates of cancer, detection of cancer at earlier stages, and smaller tumors conditional on cancer detection among compliers compared to always-takers.

Naturally, we are also interested in comparing compliers to never-takers: those who do not get screened even after the age-40 recommendation is in effect. Since the cancer status of women who do not get screened is inherently difficult (or impossible) to observe, we will draw on a clinical model of breast cancer incidence and progression to estimate the cancer profile of never-takers. Before turning to this exercise in the next section, we can use the available data to compare compliers and never-takers on various non-cancer characteristics.

Specifically, we use the discrete onset of the recommendation at age 40 in a regression discontinuity framework to implement the Abadie (2002, 2003) approach to characterizing compliers and never-takers. Figure 4 shows the results. The left-hand panel compares various characteristics of compliers and never-takers; for completeness, the right-hand panel compares compliers to always-takers. The top panel examines preventive health behaviors and prior health care use in the HCCI data. The bottom two panels examine insured women in the BRFSS data; these data allows us to observe additional health behaviors and demographic characteristics. Appendix B contains more detail on the estimation approach and also shows the average characteristics of the population and the subset who receive a mammogram, by age.

Overall, Figure 4 suggests that women who receive a mammogram as a result of the recommendation are more likely to comply with other recommended preventive care than women who do not get a mammogram even in the presence of the recommendation. In particular, both data sets indicate that compliers are more likely to get flu shots and Papanicolaou tests (also known as Pap tests, which are used to screen for cervical cancer) than never-takers. The HCCI data also indicate that compliers have lower health care spending and have fewer emergency room visits than never-takers. These results are consistent with Oster (2020)'s finding that when a health behavior is recommended, those who take it up also tend to exhibit other positive health behaviors. The results are also broadly consistent with related patterns reported by Kowalski (2019) in the context of selection into participation in clinical trials. Interestingly, however, we find no evidence of

pronounced differences between compliers and never-takers on non-healthcare dimensions; they look similar on other health behaviors (such as seat belt use and alcohol consumption) as well as on basic demographics.

III. Model and estimation

The empirical patterns documented in the preceding section indicate that the women who respond to the mammogram recommendation have a lower incidence of cancer than those who seek mammograms in the absence of a recommendation. To evaluate the implications of this selection for alternative, counterfactual timings of the screening recommendation (such as at age 45 instead of age 40), we write down a stylized model of mammogram decision making. We then estimate this model using the observed patterns shown in Section II combined with a clinical oncology model of the underlying cancer incidence in the population and tumor evolution in the absence of detection. The clinical oncology model provides the (hitherto absent) crucial information on the cancer disease burden of women who respond to the mammogram recommendation compared to women who do not. Naturally, we explore sensitivity to alternative clinical assumptions.

A. A descriptive model of mammogram choice

Consider a woman *i* in a given year she is observed in the data.¹² We model the annual decision of whether or not to have a mammogram; annual decision frequency seems natural given that mammogram screening tends not to be done more frequently than once a year. Absent any recommendation to do so, we assume that the "organic" decision to have a mammogram follows a simple probit, so that

$$\Pr(m_i^o = 1) = \Pr(\alpha^o + \gamma^o a_i + \delta_{in-situ}^o I(c_i^{in-situ}) + \delta_{invasive}^o I(c_i^{invasive}) + \varepsilon_i^o > 0), \quad (1)$$

where m_i^o is an indicator for whether woman *i* had a mammogram in that observed year, a_i is woman *i*'s age that year, $c_i = \{c_i^{in - situ}, c_i^{invasive}\}$ describes woman *i*'s undiagnosed cancer status that year, and ε_i^o is a (standard) Normally distributed error term. Following our discussion in Section II, our baseline specification summarizes cancer status c_i with two indicator variables, one that indicates an in-situ tumor and another that indicates an invasive tumor; the omitted category is no cancer.

If it is recommended that woman *i* obtain a mammogram, we model her response to the recommendation as a second, subsequent decision that is taken within the same year. That is, if a woman has already decided to have a mammogram "organically" based on equation (1), a recommendation has no additional impact. But for women who decided not to have a mammogram organically (that is, $m_i^o = 0$), a second decision point arises due to the recommendation, and we model this second decision point in a similar fashion, except that the parameters are allowed to be different:

¹²We observe women for one, two, or three years. As discussed below, this is a static model, which does not use the panel dimension, so we essentially treat the entire data as a cross-section of woman-years, each denoted by i.

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

$$\Pr(m_i^r = 1 | m_i^o = 0) = \Pr(\alpha^r + \gamma^r a_i + \delta_{in-situ}^r I(c_i^{in-situ}) + \delta_{invasive}^r I(c_i^{invasive}) + \varepsilon_i^r > 0),$$
⁽²⁾

where ε_i^r is a (standard) Normally distributed error term, drawn independently from $\varepsilon_i^{o,13}$ This model assumes that the impact of the recommendation is (weakly) monotone for all women. For each woman, it only increases the probability that she has a mammogram, a feature that seems (to us) natural.¹⁴

Since we do not directly observe whether a mammogram was taken for organic reasons or in response to a recommendation, the probability that woman *i* obtains a mammogram in the year she is observed is given by

$$\Pr(m_i = 1) = \begin{cases} \Pr(m_i^o = 1) & \text{if not recommended} \\ \Pr(m_i^o = 1) + \Pr(m_i^r = 1 | m_i^o = 0) \Pr(m_i^o = 0) & \text{if recommended} \end{cases}$$

We use the model's results to quantify the degree of selection into mammograms in the presence and absence of a recommendation, and to examine how the nature of this selection affects the impact of recommendations. To do so, we use the model estimates to predict mammogram rates and mammogram outcomes under the current recommendation to begin mammograms at age 40 as well as under a counterfactual recommendation to begin at age 45. Consistent with our focus on selection, we also examine how alternative, counterfactual selection into mammograms in response to the recommendation would change the impact of changing the recommended age of beginning mammography from 40 to 45.

Discussion.—Importantly, this is a descriptive, or statistical model of mammogram choice, rather than a behavioral one. This is most apparent from the fact that we use the cancer status c_i as an explanatory variable, when naturally this cancer status is unknown by undiagnosed women. Cancer status c_i is also unobserved by the econometrician; we describe below the clinical model of tumor evolution which we use to "fill in" these missing data, thus essentially integrating over the population distribution of this cancer status component.

We take this modeling approach for several reasons. First, many of the outcomes in this setting are difficult to assess or monetize, e.g. the stress and anxiety associated with falsepositive test results or the non-monetary costs associated with the breast cancer treatment (even if successful). This makes it difficult to translate the rich set of outcomes into a single metric of utility. Second, our key focus is on the impact of the recommendation policy. With a perfectly informed population of women, recommendations should have no impact, yet the data in Section II show a clear increase in the mammogram rate in response to the age 40 recommendation. We could try to attribute this recommendation-induced increase in

¹³While this independence assumption may appear restrictive, note that equation (2) only applies to those women who elected not to obtain an "organic" mammogram. It is therefore effectively restricted to women with "low enough" ε_{i}^{0} , so that much of the potential correlation is already conditioned out. ¹⁴That is, as in the analysis of Section II.C, we assume that there are no defiers. As will become clear later, other than appearing a

natural assumption to us, it also simplifies the intuition of how counterfactual recommendation policies play out.

mammogram rate to improved information, but this would require us to make assumptions about what type of information is being revealed and how, or why women did not have such information to begin with. We prefer instead to remain agnostic about the behavioral channel by which the recommendation affects screening rates. Finally, a descriptive model of decision making does not require us to try to reconcile observed patterns of decisions with optimal behavior, or model deviations from optimality. The drawback is, of course, that we will not be able to engage with other policy changes or with the impact of changes in the recommendation policy on individual welfare directly, but rather will only evaluate changes in recommendation policies through their effect on observed outcomes.

Another key feature of our setup is that we model the mammogram decision to be a static and perhaps naive—one. The decision is static in the sense that we assume that women do not take into account, for example, the time elapsed since their most recent mammogram (if any).¹⁵ The decision is naive in the sense that we assume that women, when deciding to get a mammogram or not, do not explicitly take into account their propensity to get a mammogram in future years. This assumption seems not unrealistic, and simplifies the model. This assumption is particularly important in the context of our counterfactual exercise, which holds the estimated model as given while we change the age at which it is recommended to begin mammography. Specifically, in considering the changes that occur when the mammogram recommendation begins at age 45 instead of 40, our static model assumes that this would have no impact on women aged 39 or younger. In a dynamic model with forward-looking agents, however, it could increase the propensity of women under age 40 to get a mammogram. Our current model could in principle capture such dynamics implicitly by allowing serial correlation in ε_i^o and in ε_i^r . However, because we have a relatively short panel, and because we only use age to match the two main data sets, it would be hard to identify such a serial correlation structure. Consistent with this being a fairly inconsequential assumption, Figure 2 shows very low rates of pre-recommendation mammograms, and no evidence that mammogram rates decline in the year or two that are right before age 40 (when forward-looking women might anticipate their future mammogram).

B. Implementation

A clinical model of tumor appearance and evolution.—To complete the empirical specification, we specify a clinical oncology model of tumor appearance and tumor evolution. The oncology model has two important roles in our analysis, one for estimation and another for our counterfactual exercises. For estimation, the key role of the oncology model is that it allows us to "impute" cancer status for the "never-takers," i.e., the women who do not get screened even when it is recommended. This clinical model delivers two key elements. First, it produces the underlying incidence of cancer (and cancer type) by age. This cannot be directly observed in data since cancer incidence is only observed conditional

¹⁵While restrictive, there is no strong evidence of such dynamic patterns in the data. We only have a short panel of at most three years for each woman, so it is difficult to apply any formal statistical testing. However, conditional on having two mammograms during the three years of mammogram claims we observe (2009–2011), the frequency of getting a mammogram "every other year" (that is, getting mammograms in 2009 and 2011 but not in 2010) is not more likely than getting a mammogram in consecutive years (34%, relative to 39% for 2009 and 2010, and 27% for 2010 and 2011).

on screening. Intuitively, since we observe the rate of cancer among those who get screened and the share of women who get screened, then, with the estimate of the overall rate of cancer from the clinical model, we can deduce the rate of cancer in the unscreened population. Second, the clinical model provides (counterfactual) predictions for the rate at which tumors would progress in the absence of detection and treatment (the so-called "natural history" of the tumor). Since breast cancer is usually treated once diagnosed, rather than being monitored without treatment, it is difficult (perhaps impossible) to directly estimate the natural history of tumors from existing data. This latter element is particularly important for our counterfactual exercises, in which the effect of different selection patterns depends on the share of cancer cases that get diagnosed, as well as how early tumors are found. In order to assess how clinically important early diagnosis is (e.g., in its effect on mortality), a model of tumor evolution is needed.

For the clinical model, we draw on an active literature creating clinical/biological models of cancer arrival and growth. Specifically, we draw on the work of the Cancer Intervention and Surveillance Modeling Network (CISNET) project funded by the National Cancer Institute to analyze the role of mammography in contributing to breast cancer mortality reductions over the last quarter of the 20th century. As part of this effort, seven different groups¹⁶ developed models of breast cancer incidence and progression (Clarke et al. 2006). For convenience, we focus on one of these models, the Erasmus model (Tan et al. 2006). As we discuss below, we also confirm that our main results are not sensitive to alternative specifications designed to produce markedly different estimates for the key objects (the underlying incidence of cancer and cancer types).

We briefly summarize the Erasmus model here; Appendix C describes the model in much more detail. Starting with a cancer-free population of 20-year-old women, the Erasmus model assumes that breast tumors appear at a given age-specific rate (that is increasing in age). When they appear, tumors are endowed with a given invasive potential and initial rate of growth, and then evolve accordingly over time with respect to those two characteristics. Tumors can either be invasive, leading to death of the women if not detected early enough, or be in-situ. In-situ tumors are not themselves harmful but may either transform into a harmful invasive tumor or remain benign. In some sense, a key issue in the debate over mammograms is the extent to which tumors that are detected early (e.g. in-situ tumors) would have become harmful if not detected or would have remained benign; Marmot et al. (2013) discusses how, depending on the method of analysis, a wide variety of estimates can be obtained when trying to answer this question. The Erasmus model further classifies tumors by whether or not they are detectable by screening, which in the case of invasive tumors depends on their size and in the case of in-situ tumors are fatal.

The original Erasmus model was calibrated using a combination of Swedish trial data and US (SEER) population data. To better match the cancer incidence rates in the SEER data

¹⁶The composition of the CISNET consortium has changed over time, but the seven groups who produced models for the original publication in 2006 were affiliated with the Dana-Farber Cancer Center, Erasmus University Rotterdam, Georgetown University Medical Center, University of Texas M.D. Anderson Cancer Center, Stanford University, University of Rochester, and University of Wisconsin-Madison.

Am Econ Rev. Author manuscript; available in PMC 2021 July 23.

(birth cohorts 1950–1975), we introduce a proportional shifter of overall cancer incidence and calibrate this parameter on the SEER data. Appendix Figure A.6 shows the calibrated model's predictions—under the assumption of no screening—of the share of women with cancer at each age, and the share of existing cancers that are in-situ (rather than invasive) by age.

Estimation and identification.—We estimate the model using method of moments. The observed moments we try to match are the mammogram screening rate at each age (Figure 1), the true positive rate at each age (Figure 2a), and the share of tumors at each age that are in-situ conditional on true positive (as in Figure 3a).¹⁷ Because identification is primarily driven by the discontinuous change in screening rates at age 40, we weight more heavily moments that are closer to age 40 than moments that are associated with younger and older ages.¹⁸

To generate the corresponding model-generated moments, we simulate a panel of women starting at age 20, and use the clinical model described above to generate cancer incidence and tumor growth for each woman. We then apply our mammogram decision model, by age and recommendation status, to each simulated woman who is alive and has yet to be diagnosed with cancer. The simulated cohort allows us to see the fraction of women with a detectable (by mammogram) tumor at each age, and thus generate the mammogram rate, and the true positive rate (by cancer type) conditional on screening. As mentioned above, for cancer type, we distinguish only between in-situ and invasive tumors.

With this simulated population of women, an assumed value of parameters associated with the mammogram decisions with and without recommendation (equations (1) and (2)) and the observed policy recommendation (40 and above), the model generates an age-specific share of women who are screened, and the tumor characteristics (in-situ and invasive rates), conditional on getting screened. We then search for the parameters that minimize the (weighted) distance between these generated moments and the observed moments described above.

Although the model is static, it does have a dynamic element because we calculate the model-generated moments only for women who were not diagnosed with cancer in previous years, and for those who did not die (from breast cancer or other causes) prior to the given age. Specifically, because the mammogram decision applies to women who have yet to be diagnosed with cancer, fitting the model requires calculating the rate of cancer among the population who is eligible to be screened, which includes those who have currently undiagnosed cancer or no cancer, but does not include those who are dead or already diagnosed. Appendix D provides more detail on this and other aspects of the estimation.

¹⁷Figure 3a shows the share of all diagnosed cancers (in the SEER data) that are in-situ, but the model produces a different metric: the share of screening mammogram-diagnosed cancers that are in-situ. Cancers that are clinically diagnosed are highly unlikely to be insitu, so the SEER value likely underestimates the true value of share in-situ for screening mammogram-diagnosed cancers. Appendix D describes how we adjust the SEER moments to account for this. ¹⁸Specifically, the weight on moments associated with ages 39 and 41 is 10/11 of the weight on the age 40 moment, the weight on

moments associated with ages 38 and 42 is 9/11 of the weight on the age 40 moment, and so on.

For our counterfactual exercises, the estimates from the mammogram choice model-and the assumption that choices would be smooth in age through age 40 in the absence of the recommendation-allow us to predict mammogram decisions and outcomes under counterfactual scenarios. Crucially, the model estimates allow us to forecast the cancer characteristics of women who (counterfactually) do not get screened and whose cancer may therefore progress in the absence of diagnosis. The key parameters are δ^o and δ^r , which capture the nature of selection into mammogram screening. Positive selection (i.e. positive δ) implies that women with cancer (or with invasive vs. in-situ cancer) are more likely to get a mammogram than are woman without cancer. A negative δ implies the opposite. Both types of selection are plausible. Positive selection could arise, for example, if women with a greater risk of breast cancer (e.g. due to family history) are more likely to get a mammogram; negative selection could arise, for example, if women with certain underlying characteristics (e.g. risk aversion) are both more likely to get a mammogram and also more likely to avoid risk factors linked to breast cancer. Importantly, by allowing δ^o and δ^r to be different, the model allows for the nature of selection to be different for organic and recommendation-driven mammograms. Identification of these selection effects is driven by comparing the share of cancer in the population (which is data provided by the clinical oncology model) to the true positive mammogram rates. The extent to which this relationship changes discretely at age 40, when the recommendation kicks in, allows us to separately identify δ^o and δ^r .

IV. The impact of alternative screening policies

A. Model fit and parameter estimates

Figure 5 presents the model fit to the key moments, which we view as quite reasonable. The parameter estimates are shown in Table 2. It may be easiest to see the implications of these parameters in the context of our counterfactual results, but one can already infer the general pattern by focusing on the four δ parameters, which indicate the extent of selection into mammogram. The two δ^{o} parameters are positive and relatively large, indicating strong positive selection into the "organic" decision to have a mammogram. For example, for the average woman-year in the sample (that is, using the distribution of ages in the sample), the estimated coefficients imply that the "organic" mammogram rates for women with either an in-situ or invasive tumor are much higher (0.30 and 0.57, respectively) relative to the "organic" mammogram rates for cancer-free women (0.20).

In contrast, the two δ^r parameters tell a different story. The estimates suggest that there is no differential selection into the "recommended" decision for women with in-situ tumors (relative to cancer-free women), and that essentially no woman with an invasive tumor selects into mammogram due to the recommendation. This result is driven by precisely the patterns in the data that identify these parameters, and which were presented in Figure 3a. Namely, conditional on diagnosis, the share of in-situ tumors rises sharply at age 40, so that virtually all the increase in detected cancers reflects in-situ tumors. As we show below, this pattern has a critical effect on our results, because women without cancer or with in-situ tumors—who constitute the primary incremental positive mammogram results may not face drastic health implications if those tumors would instead be discovered several years later.

We note that the large confidence intervals on $\delta_{invasive}^{o}$ and $\delta_{invasive}^{r}$ reflect the fact that the estimates imply that virtually all women with invasive tumors who get screened do so organically, with essentially no women with invasive tumors getting screened in response to the recommendation; as a result, the likelihood function is fairly at for high values of $\delta_{invasive}^{o}$ and low values of $\delta_{invasive}^{r}$. But for exactly the same reason, these imprecise estimates of the parameter have little impact on the counterfactual results, as reflected by the much tighter standard errors associated with the counterfactuals of interest reported in the next section.

B. Implications

We apply the estimated parameters from Table 2 to analyze outcomes under various counterfactual recommendations. For concreteness, we focus on outcomes under the current recommendation to begin mammograms at age 40 as well as under a counterfactual recommendation to begin at age 45. Our model is well suited for such a counterfactual exercise: we simply assume that mammogram decisions are based on the "organic" decision until age 45, and only at age 45 is there a second, recommendation-induced decision. Given the static nature of the model, mammogram rates will remain the same until age 40, and would be the same (conditional on cancer status) from age 45 and on, but will decrease for women aged 40-44 without a recommendation. We choose a counterfactual recommendation that begins at age 45 because this is not too far out of sample, and also in the range of realistic policy alternatives; Canada, for instance, recommends routine screening beginning at age 50 (Kadiyala and Strumpf 2011). Of course, such counterfactuals do require us to rely on our assumption of a linear age profile in order to predict outcomes for always-takers beyond age 40 in a counterfactual world in which the recommendation does not occur until age 45; while this strikes us as not unreasonable, given that the linear specification in age seems to fit the data well, it is of course an important (and untestable) assumption.

For both the age 40 and age 45 recommendations, we also examine how alternative, counterfactual selection into mammograms in response to the recommendation would change the recommendation's impact. The main outcomes we generate under the various counterfactuals are age-specific mammogram rates, mammogram outcomes (specifically, negative, false positive, and true positive, as well as tumor type), total health care spending, and mortality. We do not attempt to quantify other potential consequences of a change in recommendation (such as the opportunity to use less invasive treatments for early-stage diagnoses, or increased anxiety from false positive results, which are more uncertain (Welch and Passow 2014)).

Throughout the counterfactual exercises, mammogram rates are generated directly from the parameter estimates in Table 2, and mammogram outcomes are generated based on the parameter estimates in Table 2 and the underlying incidence and natural history of breast cancer tumors from the Erasmus model. We also use the Erasmus model's parameters in order to map detection of tumors to subsequent mortality, allowing us to translate the estimated changes in detection into implied changes in mortality. Finally, we use the auxiliary data from Figure A.2b on how health care spending varies with age and

mammogram outcomes to translate the estimated change in mammogram rates and mammogram outcomes into implied spending changes. Appendix E provides more details behind these counterfactual calculations.

Shifting the age of recommendation from 40 to 45.—Table 3 shows the implications of shifting the recommendation from age 40 to age 45, given the estimated response to recommendations from Table 2. We focus on the implications for women ages 35–50.

Panel A summarizes the implications for screening and spending; Figure 6 shows how the age profile of screening and screening outcomes change with this counterfactual. Changing the recommended age from 40 to 45 reduces the average number of mammograms a woman receives between ages 35 and 50 from 4.7 to 3.8, an almost 20 percent decline. By design, all of the "lost" mammograms occur between ages 40 and 44. Naturally, the vast majority of these "lost" mammograms would have been negative (89.5%) or false positive (10.4%). Moving the recommendation to age 45 decreases the average number of false positives a woman experiences over ages 30–45 by 0.09. The fraction of true positive mammograms that are "lost" due to the later recommendation, while small in absolute number (0.0004 per woman), is not negligible, and it constitutes an approximately 6% reduction in the cancer detection rate. Of the "lost" true positives, however, all are in-situ since our estimates imply that the recommendation effectively induces no additional women with invasive cancer to get screened. Thus, any changes in mortality are due to in-situ tumors that go unscreened and later become invasive.

The last row of Panel A shows that changing the recommendation age to 45 reduces total health care spending over ages 35–50 per woman by about \$320, or about half a percent. This reduction in spending arises from a combination of a level and composition effect. The dominant factor is naturally the decline in the overall number of mammograms. We estimate that women who have a mammogram in a given year are expected to spend approximately \$570 more (on average, averaging over ages 40–44) over the subsequent 12 months relative to women with no mammograms, and that moving the recommendation age to 45 results in 0.9 fewer mammograms per woman. This would mechanically result in approximately \$510 lower spending. The estimated spending reduction is lower (\$320) because of selection. The "lost" mammogram results are associated with, by far, the highest expected subsequent spending (see Figure A.2b). True-positive mammograms account for a larger share of mammograms in the counterfactual scenario (0.53%, relative to 0.44% under the age-40 recommendation).

Panel B documents the implications of this counterfactual for health outcomes. The lower detection rate of cancers is associated with 5 more women per 100,000 who are dead by the age of 50; all of this increase in deaths comes from increased breast cancer mortality. The results thus suggest that, relative to an age-45 recommendation, an age-40 recommendation increases spending by about \$32 million per 100,000 women (during the ages of 35–50), and prevents about 5 additional deaths by age 50 per 100,000 women; the cost per life saved is thus about \$6 million.

Naturally, these mortality implications are driven by the assumptions in the clinical oncology model, about which there is a range of views (Clarke et al. 2006; Welch and Passow 2014). In addition, our analysis considers only the costs in terms of health care spending, and does not consider the disutility of stress and anxiety created by false positives or additional medical care. For both reasons, our goal here is not to emphasize a specific estimate of the cost per life saved per se, but rather to examine whether and how this type of counterfactual policy exercise can be affected by the nature of selection into mammograms in response to the recommendation, a question we turn to in the next section.

Consequences of selection patterns in response to mammogram.—Table 4 illustrates the importance of selection in response to the recommendation. To do so, Panel A replicates the results from Table 3, while Panels B and C contrast them with what the results would be under alternative selection responses to the recommendation. Under both alternative selection models, we maintain our estimated selection associated with the "organic" mammogram decision, but vary the nature of selection into mammograms in response to the recommendation. One case (Panel B) assumes no selection, which is conceptually consistent with the idea of using estimated mammogram treatment effects from randomized experiments to inform the recommendation policy (as in, for example, Welch and Passow 2014); in practice we do this by assuming that $\delta^r = 0$.¹⁹ The other case (Panel C) assumes that selection in response to the recommendation is positive, and is the same as in the organic decision; we implement this counterfactual by assuming that δ^r is equal to our estimated δ^o .

In both counterfactual selection cases we consider, we adjust the model to maintain the same age-specific mammogram rates under a given recommendation regardless of the assumed selection, so that only the nature of selection changes; Appendix E provides more detail. By design, therefore, the mammogram rates (first row of each panel) remain almost the same across all three selection models,²⁰ and therefore the spending effect associated with each of these cases also remains almost identical (second row of each panel). In contrast, the importance of selection is shown in the third row of each panel: different patterns of selection affect the reduction in deaths from moving the recommendation to age 40 compared to age 45. For example, while our estimates that are based on observed selection imply that moving the recommendation from 45 to 40 saves 5 additional lives (by age 50) per 100,000 women, which corresponds to a cost of about \$6.3 million per life saved, random selection would imply over three times as many lives saved (18 per 100,000), corresponding to a cost of about \$1.9 million per life saved. At a more extreme case of selection, assuming that the strong positive selection associated with "organic" selection would also apply to the selection in response to the recommendation, would imply almost

¹⁹Note that here we have in mind a conceptual randomized experiment with full compliance. Of course, in practice, full compliance is rare, and the complier population to the experiment is itself not random, although it may be differentially selected from the complier population to the recommendation. In a recent paper, Kowalski (2019) argues that in practice the women most likely to receive mammograms when encouraged to do so in a randomized clinical trial are healthier, and hence benefit less from mammograms. ²⁰Although not seen in the table due to rounding, the mammogram rates are not exactly the same across the panels because the nature of selection leads to differential mortality (discussed below), which in turn (slightly) affects the set of women "eligible" for a screening mammogram.

nine times as many lives saved (45 per 100,000 women), corresponding to a cost per life saved of about \$0.86 million.

The qualitative results are intuitive. As selection associated with the recommendation is more negative (i.e. women who respond are less likely to have cancer), the recommendation for earlier mammograms is less effective in finding tumors that would have not been found otherwise or tumors that would otherwise be found only later. However, if the selection associated with the recommendation were very positive (i.e. women who respond are more likely to have cancer), an earlier recommendation would be more effective. Thus, out of the three selection scenarios considered, earlier recommendation is most beneficial if the selection response to the recommendation is the same as under organic selection, which was highly positive (Panel C). While it is not immediately clear how in practice to achieve such strong positive selection in response to the recommendation, this result suggests that better targeting of the recommended mammogram to women with higher a-priori risk of cancer could-if feasible-have dramatic effects on the mortality benefits from the recommendation.²¹ The comparison between our estimated selection (panel A) and the "no selection" case (panel B) is an intermediate case. Because we estimate negative selection for invasive tumors, an earlier recommendation is more effective (i.e. more women with cancer would be screened) under random selection, and the cost per life saved is therefore lower.

Sensitivity.—The data allow us to estimate characteristics of always-takers and compliers, and to see that compliers have a lower incidence of cancer than always-takers (see Figures 2a and 3a). However, our counterfactuals require us to also estimate the cancer status of never-takers, as well as how cancer would evolve if (counterfactually) screening occurred at a later age. For both of these endeavors, we relied heavily on the underlying natural history ("clinical") model of breast cancer. We therefore examine the sensitivity of our conclusions to changing key features of this model, such as the underlying incidence rate of cancer, the share of in-situ tumors that will become invasive if not treated, and the share of tumors that are non-malignant, i.e. have no potential to be invasive and therefore would never result in a breast cancer mortality.

This sensitivity analysis serves to highlight a point we have tried to emphasize throughout: the reader should not place much (or any) weight on our particular, quantitative estimates of the cost per life saved of recommending that mammography begin at 40 instead of at 45; these are quite sensitive to the assumptions underlying the clinical model. By contrast, the qualitative result we focus on—how the nature of the selection response to the recommendation affects any estimate of the impact of an earlier recommendation—is quite robust to alternative assumptions in the underlying clinical model. Appendix F discusses the specifics of how we implement the sensitivity analysis and presents the results in detail.

²¹The potential benefits of personalizing breast cancer screening recommendations have highlighted in the medical literature (e.g. Schousboe et al. 2011), and current breast cancer screening recommendations often differ across average-risk and high-risk women (where the latter is, e.g., women with a family history of breast cancer). But to the best of our knowledge our point about selection responses to recommendations has not been made previously. Our consistent selection model is one way of illustrating the potential gains from recommendation designs that affect take-up of mammograms based on unobservables.

V. Summary and possible policy implications

The debate over whether and when to recommend screening for a particular disease involves a host of empirical and conceptual challenges with which the existing literature has grappled, including how to estimate the "health" return to early screening, how to measure non-health benefits or costs, and how to monetize all of these factors (Humphrey et al. 2002; Nelson et al. 2009; Marmot et al. 2013; Welch and Passow 2014; Ong and Mandl 2015). We make no pretense of "resolving" these issues. Instead, we suggest an additional important and largely overlooked factor that can—and should—be considered: the nature of selection in response to the recommendation.

We illustrate this point in the specific context of the (controversial) recommendation that women should begin regular mammogram screenings at age 40. We document that this recommendation is associated with a sharp (25 percentage point) increase in mammogram rates, and that those who respond to the recommendation have substantially lower rates of cancer incidence than those who choose to get mammograms in the absence of the recommendation (i.e. before age 40). Conditional on having cancer, women who respond to the recommendation also have lower rates of the more lethal invasive cancer, relative to the less lethal in-situ cancer. These data speak directly to the relative cancer risks of women who select mammograms in the absence and presence of a recommendation. To further assess how the cancer risk of those who select mammograms when recommended compares to those who do not select mammograms even when recommended, we draw on a clinical oncology model to estimate the underlying cancer incidence in the non-screened population (since this is not directly observed). These results suggest that those who choose mammograms in the absence of a recommendation have substantially higher rates of both invasive and in-situ cancer than women who do not get screened; women who choose mammograms in response to the recommendation have similar rates of in-situ cancer to unscreened women but much lower rates of invasive cancer than unscreened women.

To illustrate the potential consequences of these selection responses to recommendations, we write down a stylized model of the mammogram decision, which depends on age, cancer status, and recommendation. We estimate this model using the observed empirical patterns combined with the clinical oncology model, the latter of which provides both the underlying incidence of cancer and the (counterfactual) tumor evolution in the absence of detection. We then apply the model to assess the implications for spending and mortality of changing the recommended age for beginning mammograms from 40 to 45. The specific numbers that we estimate will naturally be sensitive to the modeling assumptions; moreover, our estimates do not attempt to measure all of the the potential impacts of mammograms, such as stress.

Our focus instead is on the consequences of the selection response to the recommendation, which our estimates suggest are non-trivial. Specifically, we consider the impact of moving the recommended age of beginning mammography from 45 to 40, and how this varies under alternative selection responses to the recommendation. We hold the change in mammogram rates (and consequently the cost increase) from changing the recommended age constant, and show that the mortality implications from earlier recommended mammograms vary markedly with selection patterns. For example, under the observed selection pattern, the

number of lives saved by moving the recommendation from age 45 to 40 is less than a third of what it would be if those who responded to the recommendation were instead drawn at random from the population. This difference arises because we estimate that those who respond to the recommendation have much lower rates of invasive cancer. Conversely, our results also suggest that if it were feasible to target the recommendations to those with higher rates of cancer, shifting the recommendation from age 45 to 40 would save substantially more lives than either the observed selection patterns or random selection.

These findings suggest that the ongoing debates over whether and when to recommend screening for a disease should consider not only average costs and benefits from screening, but also the nature of selection associated with those who respond to the recommendation. They also suggest that future work exploring the impact of existing policy instruments or the design of potential new ones should consider not just aggregate impacts on mammography rates, but also the cancer incidence for compliers.

While our empirical focus has been on recommendations, these are of course only one part of a broader set of policy efforts that have been deployed or discussed for increasing disease screening. In the case of mammograms, another widely-used instrument has been lowering the financial costs of screenings. For example, in 1991 the federal government launched the National Breast and Cervical Cancer Early Detection Program to provide free screenings to women below 250 percent of the federal poverty line (Lee et al. 2014). In the same year, Medicare expanded its coverage to include bi-annual screening mammograms; subsequently, in 1998, Medicare expanded coverage further to include annual screening mammograms and to waive the deductible (O'Sullivan et al. 1997; Kelaher and Stellman 2000; Habermann et al. 2007). On the private insurance side, a number of states have mandated that insurance plans must cover mammography (Bitler and Carpenter 2016). Beyond these financial levers, there are also policy efforts to reduce non-financial barriers to mammograms. These include, for example, increasing ease of access to mammograms through programs such as mobile mammography clinics (Vang, Margolies, and Jandorf 2018), and outreach efforts designed to educate women about the benefits of mammograms and informing them of the services available to them (Levano et al. 2014).

Related to these efforts, an existing literature has studied the impact of various policy instruments on mammography rates. It has found, for example, that lowering out-of-pocket financial costs increases mammogram rates (Kelaher and Stellman 2000; Habermann et al. 2007; Finkelstein et al. 2012; Fedewa et al. 2015; Mehta et al. 2015; Bitler and Carpenter 2016; Cooper et al. 2017; Kim and Lee 2017), while increasing the distance a woman must travel to get a mammogram decreases mammogram rates (Lu and Slusky 2016). Only a few of these studies have examined differential responses to the policy by underlying health characteristics. This existing work suggests that, like our findings on the response to guidelines, those who get mammograms in response to a lower price and those who comply with assignment to mammogram treatment in a clinical trial tend to be healthier than nevertakers and always-takers (Bitler and Carpenter 2016; Kim and Lee 2017; Kowalski 2019). While one of course must be careful in generalizing too much from a few studies, our read of this existing literature is that these alternative interventions are not obviously better

targeted than recommendations in terms of the compliers, at least in the context of mammograms.

The combined evidence therefore highlights the importance of trying to better target the existing instruments. This is challenging since underlying cancer incidence, tumor stage, and tumor size are not observable without screening. However, our descriptive analyses in Section II—comparing compliers to never-takers on a host of observable characteristics—suggest that never-takers are also less likely than compliers to engage in other recommended health behaviors, such as flu shots and Pap tests. This finding is consistent with the idea that those who comply with recommendations tend to exhibit other positive health behaviors (Oster 2020). It also suggests that coordinated efforts, which attempt to draw in women who otherwise would not engage in any preventive health behaviors, could be high-value. If we are willing to extrapolate our qualitative results from breast cancer to these related contexts, our findings suggest that trying to get such women to undertake a slew of recommended health behaviors might be well-targeted at reaching women at higher risk of not only breast cancer, but perhaps also cervical cancer and the flu.

Recent analyses by clinical researchers also suggest other observables that might be useful in targeting mammograms to higher-risk groups, instead of (or in conjunction with) agebased screening recommendations. For example, Evans et al. (2019) describe the results of a randomized trial that begins regular mammograms at age 34 for women with a mother or sister who has been diagnosed with breast cancer, and an ongoing trial is investigating the impact of risk-based screening relative to standard annual screening (Esserman et al. 2017). Motivated by such work, researchers have proposed that the recommended age of beginning mammography should be based on individual risk factors such as age of first birth, number of children, and breast density (Evans, Howell, and Howell 2020; Mukama et al. 2020). Our findings underscore the potential value of such targeting, given that compliance with the existing recommendation is only about one-third, and compliers appear to be disproportionately low-risk for cancer. They also suggest the importance of analyzing the impact of targeted instruments, not only for recommendations but also for price subsidies and other policy instruments.

More broadly, our findings suggest that considering and improving selection into screening is a first-order factor in an effective design and analysis of interventions to increase screenings. However, the extent to which we can generalize our findings in this paper, in the context of mammograms, to other types of screening and preventive medicine remains an open question. The controversy surrounding the recommendation that mammography start at age 40 may generate stronger selection than in other, less controversial settings (e.g., flu shots). Whether this is true or not is an important question that we leave for future work.

Supplementary Material

Refer to Web version on PubMed Central for supplementary material.

Acknowledgments

We are grateful to Leila Agha, Emily Oster, Doug Owens, three anonymous referees, and participants in the Dartmouth/NIA P01 Research Meeting and the NBER Health Care Summer Institute for helpful comments, and to the Laura and John Arnold Foundation for financial support. This material is based upon work supported by the National Institute on Aging through Grant Number T32-AG000186 and the National Science Foundation Graduate Fellowship Program under Grant Number 1122374 (Oostrom). The authors acknowledge the assistance of the Health Care Cost Institute (HCCI) and its data contributors, Aetna, Humana, and UnitedHealthcare, in providing the claims data analyzed in this study. This study also used the linked SEER-Medicare database. The interpretation and reporting of these data are the sole responsibility of the authors. The authors acknowledge the efforts of the National Cancer Institute; the Office of Research, Development and Information, CMS; Information Management Services (IMS), Inc.; and the Surveillance, Epidemiology, and End Results (SEER) Program tumor registries in the creation of the SEER-Medicare database.

References

- Abadie Alberto. 2002. "Bootstrap Tests for Distributional Treatment Effects in Instrumental Variable Models." Journal of the American Statistical Association 97 (457): 284–92.
- Abadie Alberto. 2003. "Semiparametric Instrumental Variable Estimation of Treatment Response Models." Journal of Econometrics 113 (2): 231–63.
- Abaluck Jason, Agha Leila, Kabrhel Chris, Raja Ali, and Venkatesh Arjun. 2016. "The Determinants of Productivity in Medical Testing: Intensity and Allocation of Care." American Economic Review 106 (12): 3730–64.
- Alexander FE, Anderson TJ, Brown HK, Forrest APM, Hepburn W, Kirkpatrick AE, Muir BB, Prescott RJ, and Smith A. 1999. "14 Years of Follow-Up from the Edinburgh Randomised Trial of Breastcancer Screening." Lancet 353 (9168): 1903–08. [PubMed: 10371567]
- American Cancer Society. 2017a. "Breast Cancer Facts & Figures 2017–2018." American Cancer Society, Inc. https://www.cancer.org/content/dam/cancer-org/research/cancer-facts-and-statistics/ breast-cancer-facts-and-gures/breast-cancer-facts-and-gures-2017-2018.pdf.
- American Cancer Society. 2017b. "Limitations of Mammograms." https://www.cancer.org/cancer/ breast-cancer/screening-tests-and-early-detection/mammograms/limitations-of-mammograms.html.
- American Cancer Society. 2018. "History of ACS Recommendations for the Early Detection of Cancer in People Without Symptoms. https://www.cancer.org/health-care-professionals/american-cancer-society-prevention-early-detection-guidelines/overview/chronological-history-of-acs-recommendations.html.
- Angrist Joshua D., Imbens Guido W., and Rubin Donald B.. 1996. "Identification of Causal Effects Using Instrumental Variables." Journal of the American Statistical Association 91 (434): 444–65.
- Anwar Shamena, and Fang Hanming. 2006. "An Alternative Test of Racial Prejudice in Motor Vehicle Searches: Theory and Evidence." American Economic Review 96 (1): 127–51.
- Berry Donald A. 2013. "Breast Cancer Screening: Controversy of Impact." Breast 22: S73–76. [PubMed: 24074796]
- Bitler Marianne P., and Carpenter Christian S.. 2016. "Health Insurance Mandates, Mammography, and Breast Cancer Diagnoses." American Economic Journal: Economic Policy 8 (3): 39–68. [PubMed: 29527253]
- Bjurstam Nils, Lena Björneld Jane Warwick, Sala Evis, Duffy Stephen W, Nyström Lennarth, Walker Neil, Cahlin Erling, Eriksson Olof, Hafstr m Lars-Olof, Lingaas Halvard, Mattsson Jan, Persson Stellan, Rudenstam Carl-Magnus, Salander H kan, Säve-Söderbergh Johan, and Wahlin Torkel. 2003. "The Gothenburg Breast Screening Trial." Cancer, 97 (10): 2387–96. [PubMed: 12733136]
- Bleyer Archie, and Welch H. Gilbert. 2012. "Effect of Three Decades of Screening Mammography on Breast-Cancer Incidence." New England Journal of Medicine 367 (21): 1998–2005.
- Block Lauren D., Jarlenski Marian P., Wu Albert W., and Bennett Wendy L.. 2013. "Mammography Use Among Women Ages 40–49 After the 2009 U.S. Preventive Services Task Force Recommendation." Journal of General Internal Medicine 28 (11): 1447–53. [PubMed: 23674077]
- Blustein Jan. 1995. "Medicare Coverage, Supplemental Insurance, and the Use of Mammography by Older Women." New England Journal of Medicine 332 (17): 1138–43.

- Brett J, Bankhead C, Henderson B, Watson E, and Austoker J. 2005. The Psychological Impact of Mammographic Screening. A Systematic Review." Psycho-Oncology 14 (11): 917–38. [PubMed: 15786514]
- Clarke Lauren D., Plevritis Sylvia K., Boer Rob, Cronin Kathleen A., and Feuer Eric J.. 2006. "A Comparative Review of CISNET Breast Models Used To Analyze U.S. Breast Cancer Incidence and Mortality Trends." Journal of the National Cancer Institute, Monographs (36): 96–105. [PubMed: 17032899]
- Cooper Gregory S., Tzuyung Doug Kou Avi Dor, Koroukian Siran M., and Schluchter Mark D.. 2017. "Cancer Preventive Services, Socioeconomic Status, and the Affordable Care Act." Cancer 123 (9): 1585–89. [PubMed: 28067955]
- Cronin Kathleen A., Miglioretti Diana L., Krapcho Martin, Yu Binbing, Geller Berta M., Carney Patricia A., Onega Tracy, Feuer Eric J., Breen Nancy, and Ballard-Barbash Rachel. 2009. "CEBP Focus on Cancer Surveillance: Bias Associated with Self-Report of Prior Screening Mammography." Cancer Epidemiology, Biomarkers, and Prevention 18 (6): 1699–1705.
- Cutler David M. 2008. "Are We Finally Winning theWar on Cancer?" Journal of Economic Perspectives 22 (4): 3–26.
- Einav Liran, Finkelstein Amy, Ryan Stephen, Schrimpf Paul, and Cullen Mark R. 2013. "Selection on Moral Hazard in Health Insurance." American Economic Review 103 (1): 178–219.
- Elmore Joann G. 2016. "Solving the Problem of Overdiagnosis." New England Journal of Medicine 375 (15): 1483–86.
- Esserman Laura J., Shieh Yiwey, and Thompson Ian. 2009. "Rethinking Screening for Breast Cancer and Prostate Cancer." Journal of the American Medical Association 302 (15): 1685–92. [PubMed: 19843904]
- Esserman Laura J., and the WISDOM Study and Athena Investigators. 2017. "The WISDOM Study: Breaking the Deadlock in the Breast Cancer Screening Debate." Breast Cancer 3 (1): 1–7. [PubMed: 28232952]
- Evans D. Gareth, Howell Sacha J., and Howell Anthony. 2020. "New Evidence Confirms that Reproductive Risk Factors Can be Used to Stratify Breast Cancer Risks: Implications for a New Population Screening Paradigm." European Journal of Cancer 124: 204–06. [PubMed: 31784063]
- Evans D.Gareth, Thomas S, Caunt J, Burch A, Brentnall AR, Roberts L, Howell A, Wilson M, Fox R, Hillier S, Sibbering DM, Moss S, Wallis MG, Eccles DM, FH02 study group, and Duffy S. 2019.
 "Final Results of the Prospective FH02 Mammographic Surveillance Study of Women Aged 35–39 at Increased Familial Risk of Breast Cancer." EClinicalMedicine 7, 39–46. [PubMed: 31008449]
- Fedewa Stacey A., Goodman Michael, Flanders W. Dana, Han Xuesong, Smith Robert A., Ward Elizabeth M., Doubeni Chyke A., Sauer Ann Goding, Jemal Ahmedin2015. "Elimination of Cost-Sharing and Receipt of Screening for Colorectal and Breast Cancer." Cancer 121 (18): 3272–80. [PubMed: 26042576]
- Finkelstein Amy, Taubman Sarah, Wright Bill, Bernstein Mira, Gruber Jonathan, Newhouse Joseph P., Allen Heidi, Baicker Katherine, and the Oregon Health Study Group. 2012. "The Oregon Health Insurance Experiment: Evidence from the First Year." Quarterly Journal of Economics 127 (3): 1057–1106.
- Habbema JD, van Oortmarssen GJ, van Putten DJ, Lubbe JT, and van der Maas PJ. 1986. "Age-Specific Reduction in Breast Cancer Mortality by Screening: An Analysis of the Results of the Health Insurance Plan of Greater New York Study." Journal of the National Cancer Institute 77 (2): 317–20. [PubMed: 3461193]
- Habermann Elizabeth B., Virnig Beth A., Riley Gerald F., and Baxter Nancy N.. 2007. "The Impact of a Change in Medicare Reimbursement Policy and HEDIS Measures on Stage at Diagnosis among Medicare HMO and Fee-For-Service Female Breast Cancer Patients." Medical Care 45 (8): 761– 66. [PubMed: 17667310]
- Harding Charles, Pompei Francesco, Burmistrov Dmitriy, Welch H. Gilbert, Abebe Rediet, and Wilson Richard. 2015. "Breast Cancer Screening, Incidence, and Mortality Across US Counties." JAMA Internal Medicine 175 (9): 1483–89. [PubMed: 26147578]

- HCCI (Health Care Cost Institute). 2012. Health Care Cost and Utilization Report: 2011." https:// healthcostinstitute.org/annual-reports/2011-health-care-cost-and-utilization-report.
- Hubbard Rebecca A., Kerlikowske Karla, Flowers Chris I., Yankaskas Bonnie C., Zhu Weiwei, and Miglioretti Diana L.. 2011. "Cumulative Probability of False-Positive Recall or Biopsy Recommendation After 10 Years of Screening Mammography: A Cohort Study." Annals of Internal Medicine 155 (8): 481–92. [PubMed: 22007042]
- Humphrey Linda L., Helfand Mark, Chan Benjamin K.S., and Woolf Steven H.. 2002. "Breast Cancer Screening: a Summary of the Evidence for the U.S. Preventive Services Task Force." Annals of Internal Medicine 137 (5 Part 1): 347–60. [PubMed: 12204020]
- Jacobson Mireille, and Kadiyala Srikanth. 2017. "When Guidelines Conflict: A Case Study of Mammography Screening Initiation in the 1990s." Women's Health Issues 27 (6): 692–99. [PubMed: 28935360]
- Jørgensen Karsten Juhl, and Gøtzsche Peter C.. 2009. "Overdiagnosis in Publicly Organised Mammography Screening Programmes: Systematic Review of Incidence Trends." BMJ 339: b2587. [PubMed: 19589821]
- Jørgensen Karsten Juhl, Gøtzsche Peter C., Kalager Mette, and Zahl Per-Henrik. 2017. "Breast Cancer Screening in Denmark." Annals of Internal Medicine 167 (7): 524.
- Kadiyala Srikanth, and Strumpf Erin C. 2011. "Are United States and Canadian Cancer Screening Rates Consistent with Guideline Information Regarding the Age of Screening Initiation?" International Journal for Quality in Health Care 23 (6): 611–20. [PubMed: 21890706]
- Kadiyala Srikanth, and Strumpf Erin C. 2016. "How Effective is Population-Based Cancer Screening? Regression Discontinuity Estimates from the US Guideline Screening Initiation Ages." Forum for Health Economics & Policy 19 (1): 87–139. [PubMed: 31419895]
- Kelaher M, and Stellman JM. 2000. "The Impact of Medicare Funding on the Use of Mammography among Older Women: Implications for Improving Access to Screening." Preventive Medicine 31 (6): 658–64. [PubMed: 11133332]
- Kim Hyuncheol Bryant, and Lee Sun-Mi. 2017. "When Public Health Intervention is not Successful: Cost Sharing, Crowd-Out, and Selection in Korea's National Cancer Screening Program." Journal of Health Economics 53: 100–16. [PubMed: 28340393]
- Kolata Gina. 2009."Get a Mammogram. No Don t. Repeat." New York Times. https:// www.nytimes.com/2009/11/22/weekinreview/22kolata.html
- Kowalski Amanda E. 2019: Behavior within a Clinical Trial and Implications for Mammography Guidelines. Working Paper No. 25049, National Bureau of Economic Research.
- Lee Nancy C., Wong Faye L., Jamison Patricia M., Jones Sandra F., Galaska Louise, Brady Kevin T., Wethers Barbara, and Stokes-Townsend George-Ann. 2014. "Implementation of the National Breast and Cervical Cancer Early Detection Program: The beginning." Cancer 120 (S16): 2540– 48. [PubMed: 25099896]
- Levano Whitney, Miller Jacqueline W., Leonard Banning, Bellick Linda, Crane Barbara E., Kennedy Stephenie K., Haslage Natalie M., Hammond Whitney, and Tharpe Felicia S.. 2014. "Public Education and Targeted Outreach to Underserved Women Through the National Breast and Cervical Cancer Early Detection Program." Cancer 120 (S16): 2591–96. [PubMed: 25099902]
- Lu Yao, and Slusky David J.G.. 2016. "The Impact of Women's Health Clinic Closures on Preventive Care." American Economic Journal: Applied Economics 8 (3): 100–24.
- Maciosek Michael V., Coffield Ashley B., Flottemesch Thomas J., Edwards Nichol M., and Solberg Leif I.. 2010. "Greater Use of Preventive Services in U.S. Health Care Could Save Lives at Little or No Cost." Health Affairs 29 (9): 1656–60. [PubMed: 20820022]
- Marmot MG, Altman DG, Cameron DA, Dewar JA, Thompson SG, Wilcox M, and The Independent UK Panel on Breast Cancer Screening. 2013. "The Benefits and Harms of Breast Cancer Screening: An Independent Review." British Journal of Cancer 108 (11): 2205–40. [PubMed: 23744281]
- Mehta Shivan J., Polsky Daniel, Zhu Jingsan, Lewis James D., Kolstad Jonathan T., Loewenstein George, and Volpp Kevin G.. 2015. "ACA Mandated Elimination of Cost Sharing for Preventive Screening Has Had Limited Early Impact." American Journal of Managed Care 21 (7): 511–17.

- Miller Anthony B., To Teresa, Baines Cornelia J., Wall Claus. 2000. "The Canadian National Breast Screening Study-2: 13-Year Results of a Randomized Trial in Women Aged 50–59 Years." Journal of the National Cancer Institute 92 (18): 1490–99. [PubMed: 10995804]
- Miller Anthony B., To Teresa, Baines Cornelia J., Wall Claus. 2002. "The Canadian National Breast Screening Study-1: Breast Cancer Mortality after 11 to 16 Years of Follow-up: A Randomized Screening Trial of Mammography in Women Age 40 to 49 Years." Annals of Internal Medicine 137 (5 Part 1): 305. [PubMed: 12204013]
- Moss Sue M., Cuckle Howard, Evans Andy, Johns Louise, Waller Michael, Bobrow Lynda, and Trial Management Group. 2006. "Effect of Mammographic Screening from Age 40 Years on Breast Cancer Mortality at 10 Years' Follow-Up: A Randomised Controlled Trial." Lancet 368 (9552): 2053–60. [PubMed: 17161727]
- Mukama Trasias, Fallahad Mahdi, Tian Yu, Sundquist Kristina, Sundquist Jan, Brenner Hermann, and Kharazmi Elham. 2020. "Risk-Tailored Starting Age of Breast Cancer Screening Based on Women's Reproductive Profile: A Nationwide Cohort Study." European Journal of Cancer 124: 207–13. [PubMed: 31761537]
- Nelson Heidi D., Tyne Kari, Naik Arpana, Bougatsos Christina, Chan Benjamin, Nygren Peggy, and Humphrey Linda. 2009. "Screening for Breast Cancer: Systematic Evidence Review Update for the U. S. Preventive Services Task Force." Annals of Internal Medicine 151 (10): 727–W242. [PubMed: 19920273]
- Nyström Lennarth, Andersson Ingvar, Bjurstam Nils, Frisell Jan, Nordenskjöld Bo, and Rutqvist Lars Erik. 2002. "Long-Term Effects of Mammography Screening: Updated Overview of the Swedish Randomised Trials." Lancet 359 (9310): 909–19. [PubMed: 11918907]
- Oeffinger Kevin C., Fontham Elizabeth T. H., Etzioni Ruth, Herzig Abbe, Michaelson James S., Tina Shih Ya-Chen, Walter Louise C., Church Timothy R., Flowers Christopher R., LaMonte Samuel J., Wolf Andrew M. D., DeSantis Carol, Lortet-Tieulent Joannie, Andrews Kimberly, Manassaram-Baptiste Deana, Saslow Debbie, Smith Robert A., Brawley Otis W., and Wender Richard. 2015. "Breast Cancer Screening for Women at Average Risk: 2015 Guideline Update from the American Cancer Society." Journal of the American Medical Association 314 (15): 1599–1614. [PubMed: 26501536]
- Ong Mei-Sing, and Mandl Kenneth D.. 2015. "National Expenditure for False-Positive Mammograms and Breast Cancer Overdiagnoses Estimated at \$4 Billion a Year." Health Affairs 34 (4): 576–83. [PubMed: 25847639]
- Oster Emily. 2020. "Health Recommendations and Selection in Health Behaviors." American Economic Review: Insights 2 (2): 143–60.
- O'Sullivan Jennifer, Franco Celinda, Fuchs Beth C., Lyke Bob, Price Richard, and Swendiman Kathleen S.. 1997. "Medicare Provisions in the Balanced Budget Act of 1997. CRS Report for Congress 97–802. https://www.everycrsreport.com/files/ 19970818 97-802 0347a1fe70af1b6bbce038af2c217186942cd7cc.pdf.
- Persico Nicola. 2009: "Racial Profiling? Detecting Bias Using Statistical Evidence." Annual Review of Economics 1 (1): 229–54.
- Saad Lydia. 2009. "Women Disagree with New Mammogram Advice." Gallup http://news.gallup.com/ poll/124463/women-disagree-new-mammogram-advice.aspx.
- Schousboe John T., Kerlikowske Karla, Loh Andrew, and Cummings Steven R.. 2011. "Personalizing Mammography by Breast Density and Other Risk Factors for Breast Cancer: Analysis of Health Benefits and Cost-Effectiveness." Annals of Internal Medicine 155 (1): 10–20. [PubMed: 21727289]
- SEER (Surveillance, Epidemiology, and End Results Program). 2019. "SEER Incidence Data, 1973–2015." https://seer.cancer.gov/data/index.html.
- Segel Joel E., Balkrishnan Rajesh, and Hirth Richard A.. 2017. "The Effect of False-Positive Mammograms on Antidepressant and Anxiolytic Initiation." Medical Care 55 (8): 752–58. [PubMed: 28650923]
- Susan G Komen Foundation. 2018: "Accuracy of Mammograms." https://ww5.komen.org/ BreastCancer/AccuracyofMammograms.html.

- Tan Sita Y.G.L., van Oortmarssen Gerrit J., de Koning Harry J., Boer Rob, Habbema J. Dik F. 2006. "The MISCAN-Fadia Continuous Tumor Growth Model for Breast Cancer.". Journal of the National Cancer Institute, Monographs 36: 56–65.
- Taubes Gary.. 1997. "The Breast-Screening Brawl." Science 275 (5303): 1056–59. [PubMed: 9054004]
- Vang Suzanne, Margolies Laurie R., and Jandorf Lina. 2018. "Mobile Mammography Participation Among Medically Underserved Women: A Systematic Review." Preventing Chronic Disease 15: E140. [PubMed: 30447104]
- Welch H. Gilbert. 2015. Less Medicine, More Health: 7 Assumptions That Drive Too Much Medical Care. Beacon Press.
- Welch H. Gilbert, and Black William C.. 2010. "Overdiagnosis in Cancer." Journal of the National Cancer Institute 102 (9): 605–13. [PubMed: 20413742]
- Welch H. Gilbert, and Passow Honor J.. 2014. "Quantifying the Benefits and Harms of Screening Mammography." JAMA Internal Medicine 174 (3): 448–54. [PubMed: 24380095]
- Welch H. Gilbert, Prorok Philip C., O'Malley A. James, and Kramer Barnett S.. 2016. "Breast-Cancer Tumor Size, Overdiagnosis, and Mammography Screening Effectiveness.".New England Journal of Medicine 375 (15): 1438–47.
- Welch H. Gilbert, Schwartz Lisa M., and Woloshin Steven. 2011. Overdiagnosed: Making People Sick in the Pursuit of Health. Beacon Press.
- Zackrisson Sophia, Andersson Ingvar, Janzon Lars, Manjer Jonas, and Garne Jens Peter. 2006. "Rate of Over-Diagnosis of Breast Cancer 15 Years After End of Malm Mammographic Screening Trial: Follow-Up Study." BMJ 332 (7543): 689–92. [PubMed: 16517548]
- Zahl Per-Henrik, Maehlen Jan, and Welch H. Gilbert. 2008. "The Natural History of Invasive Breast Cancers Detected by Screening Mammography." Archives of Internal Medicine 168 (21), 2311– 16. [PubMed: 19029493]



Figure 1: Mammogram rates by age

Figure shows share of women who had a mammogram by age, from insurance claims data on a set of privately insured woman-years from 2008–2012, for mammograms between 2009–2011. Because we observe birth year, age is measured as of the start of the calendar year. Thus the mammogram rate at age 40 is the share of women who got a mammogram in the year they turned 40. Error bars (small, and therefore not visible in the figure) reflect 95% confidence intervals. N = 7,373,302 woman-years.









Figure 2: Mammogram outcomes by age

Sample is limited to the set of privately insured woman-years from the private insurance claims data who had a mammogram. N = 7,373,302 woman-years. For each age (measured by the age at the beginning of the calendar year), panel (a) shows the share of mammograms that are true positive (left hand axis) and false positive (right hand axis); the omitted category is mammograms that are negative. Panel (b) presents the share of women with a false positive by age; this reflects both mammogram rates by age from Figure 1, and the share of mammograms with a false positive by age from panel (a). Error bars reflect 95% confidence intervals.



Figure 3: Tumor characteristics and mortality by age

Panel (a) shows diagnosed breast cancer tumors by age in the SEER data from 2000–2015; N = 197,956 breast cancer diagnoses. Primary y-axis shows share of breast cancer tumors that are in-situ; secondary y-axis shows average size of diagnosed tumors. Panel (b) shows 5-year mortality for diagnosed breast cancer tumors separately by age of diagnoses and by tumor stage (in-situ and invasive) in the SEER data from 2000–2010 to account for five-year mortality outcomes by 2015; N = 147,243 diagnoses with non-missing 5-year mortality. Error bars reflect 95% confidence intervals.



Figure 4: Characteristics of who selects into mammograms

Figure reports the ratio of health care use, behavior, and demographics for compliers relative to always-takers (left panel) and compliers relative to never-takers (right panel). The mean characteristics for these groups were calculated using regression coefficients from the estimation of equation (A.1) as described in Appendix B. Error bars represent 95% confidence intervals. Standard errors are constructed using a bootstrap with 100 repetitions clustered at the age level. The error bars for "Use Oral Contraceptive Pill" and "# Drinks in Prior Month" in the right panel are truncated at zero and two for scaling; the actual bootstrap confidence intervals are larger. The sample in the first section is a set of privately insured woman-years from HCCI from 2008–2012, for mammograms between 2009–2011. The sample in the second and third sections is from BRFSS for even years 2000–2012, restricted to women with any health insurance (the data do not distinguish between public or private insurance status). Details for each outcome are listed in Appendix Figures A.3, A.4, and A.5.



Figure 5: Model fit

Figure shows model fit by comparing the observed patterns of mammogram rates, outcomes, and types of diagnoses by age to the fitted values from the model based on the parameter estimates from Table 2. The observed data on mammograms (Panel (a)) was previously shown in Figure 1; the observed data on share of mammograms that are true positives was previously shown in Figure 2a; the observed data on the share of diagnoses that are in-situ is a modified version of the data shown in Figure 3a. While Figure 3a presented the share of all

diagnosed cancers that are in-situ, we match the share of mammogram-diagnosed cancers that are in-situ, as shown in Panel (c). Appendix D provides more detail.



Figure 6: Impact of changing the mammogram recommendation age from 40 to 45, by age Figure reports the model predictions - by age - for mammogram rates, mammogram outcomes, and the share of diagnoses that are in-situ, based on the parameter estimates from Table 2. As in Table 3, we report the model predictions both under the status quo recommendation that mammograms begin at age 40 and the counterfactual recommendation that mammograms begin at age 45.

_

Table 1:

Summary statistics

	No.of Observations		Health Care Spending (\$US)	
	N (000s)	Share	Total	Out-of-pocket
No mammogram	5,166.2	0.701	4,300	625
Mammogram	2,206.9	0.299	4,985	751
Conditional on mammogram:				
Negative	1,977.8	0.896	4,552	715
False positive	214.6	0.097	6,106	952
True Positive	14.4	0.007	47,639	2,821

Table shows summary statistics from insurance claims data on a set of 35–50 year old privately insured women from 2008–2012, for mammograms between 2009–2011. Each observation is a woman-year. 12-month spending measures health care spending in the 12 months after the mammogram (including the mammogram itself) for those with a mammogram. For those without a mammogram, we draw a reference date from the distribution of actual mammograms in that year. All reference dates are set to be the first of the given month. Spending is measured in the 12 months after this reference date.

Table 2:

Parameter estimates

Parameter	Estimate	95% Confidence Interval
ao	-5.21	[-5.63, -4.48]
γ^{o}	0.10	[0.08, 0.11]
$\delta^o{}_{in\text{-situ}}$	0.36	[0.29, 0.97]
$\delta^o{}_{invasive}$	1.13	[0.98,56.73]
a ^r	0.29	[-0.63, 1.18]
$\gamma^{\rm r}$	-0.03	[-0.05, 0.00]
$\delta^r_{\text{in-situ}}$	-0.01	[-0.20, 0.77]
$\delta^r_{\text{invasive}}$	-4.67	[-143, -0.01]

Table shows the parameter estimates from the mammogram decision model. Confidence intervals are calculated using 100 repetitions of the bootstrap.

Table 3:

Impact of changing the mammogram recommendation age from 40 to 45

	Rec at Age 40	Rec at Age 45	Change			
A. Screening and spending (per woman)						
Mammograms	4.70 (0.06)	3.80 (0.14)	-0.90 (0.08)			
Negative	4.22 (0.05)	3.42 (0.12)	-0.81 (0.07)			
False positives	0.46 (0.01)	0.36 (0.02)	-0.09 (0.01)			
True positives	0.0208 (0.0024)	0.0204 (0.0024)	-0.0004 (0.0001)			
In-situ diagnoses	0.0063 (0.0005)	0.0060 (0.0005)	-0.0004 (0.0001)			
Invasive diagnoses	0.0145 (0.0019)	0.0145 (0.0019)	0.0000 (0.0001)			
Total health care spending (\$)	71,326 (128)	71,007 (155)	-319 (29)			
B. Mortality (per 1,000 women by age 50)						
Dead	15.98 (0.53)	16.03 (0.53)	0.05 (0.03)			
Dead from breast cancer	8.23 (0.53)	8.28 (0.53)	0.05 (0.03)			
Dead from other reason	7.75 (0.00)	7.75 (0.00)	0.00 (0.00)			
Years alive, per woman	15.87 (0.00)	15.87 (0.00)	-0.0002 (0.0001)			

Table reports model predictions for various outcomes under the status quo recommendation that mammograms begin at age 40 (column 1) and the counterfactual recommendation that mammograms begin at age 45 (column 2). The predictions are generated using the parameter estimates from Table 2, and simulated women's life histories under a non-screening regime based on the clinical oncology model. Panel A reports the average number of mammograms and different mammogram outcomes per woman over ages 35–50. Panel B shows the share of women dead (and from different causes) by age 50, as well as the number of years alive on average between 35 and 50. Standard errors are calculated using 100 repetitions of the bootstrap.

Table 4:

Spending differences for different components of spending

	Recomme	D. 66	
	Age 40	Age 45	Difference
A. Estimated Selection			
Mammograms (per woman)	4.70 (0.06)	3.80 (0.14)	-0.90 (0.08)
Total health care spending (\$ per woman)	71,326 (128)	71,007 (155)	-319 (29)
Dead by age 50 (per 1,000 women)	15.98 (0.53)	16.03 (0.53)	0.05 (0.03)
B. No Selection			
Mammograms (per woman)	4.70 (0.06)	3.80 (0.14)	-0.90 (0.08)
Total health care spending (\$ per woman)	71,364 (111)	71,024 (147)	-340 (37)
Dead by age 50 (per 1,000 women)	15.84 (0.47)	16.02 (0.53)	0.18 (0.06)
C. Consistent Selection			
Mammograms (per woman)	4.70 (0.06)	3.80 (0.14)	-0.90 (0.08)
Total health care spending (\$ per woman)	71,450 (87)	71,068 (134)	-382 (48)
Dead by age 50 (per 1,000 women)	15.54 (0.39)	15.99 (0.52)	0.45 (0.13)

Table reports model predictions under the status quo recommendation that mammograms begin at age 40 (column 1) and the counterfactual recommendation that mammograms begin at age 45 (column 2). Each panel reports results under different assumptions about the nature of selection both in the absence and presence of a recommendation. Panel A reports results based on the estimated selection patterns; these results repeat

findings shown previously in Table 3. Panel B repeats the same exercises as in Panel A, but instead of using the estimated selection (i.e. the δ^{O} and δ^{r} parameters shown in Table 2), we instead assume "no selection" (i.e. we set $\delta^{O} = \delta^{r} = 0$). Panel C also repeats the exercises in Panel A but now assumes "consistent selection" (i.e. we set δ^{r} equal to our estimates of δ^{O} in Table 2). In both Panel B and C, we hold the overall mammogram rate fixed at Panel A's predicted age-specific mammogram rates (which of course varies in column 1 and column 2), so that the counterfactuals across panels consider differences in selection, not in levels. To do this we adjust the intercept a_{r} for each age and counterfactual to match the age-specific mammogram rates in Panel A, assuming the simulated life histories and cancer status remains constant. The small differences in mammograms in Panel A and Panel C are due to changes in the denominator of simulated life histories. Specifically, since fewer women die in Panel C, there are more years where they could potentially obtain a mammogram. Standard errors are calculated using 100 repetitions of the bootstrap.